Author's response to reviews

Title: Does burnout among doctors affect their involvement in patients' mental health problems? A study of videotaped consultations

Authors:

Else M Zantinge (e.zantinge@nivel.nl)
Peter FM Verhaak (p.verhaak@nivel.nl)
Dinny H de Bakker (d.debakker@nivel.nl)
Klaas van der Meer (klaas.van.der.meer@med.umcg.nl)
Jozien M Bensing (j.bensing@nivel.nl)

Version: 4 Date: 19 March 2009

Author's response to reviews: see over
To: BMC Family Practice
   The Editor

Utrecht, 19 March 2009

Dear Editor,

Please find attached our revised paper ‘Does burnout among doctors affect their involvement in patients’ mental health problems? A study of videotaped consultations’, which we would like to resubmit for publication in BMC Family Practice. We have added a sentence about the ethics of our study in the method section (subheading ‘subjects’). Furthermore we checked if our paper conforms to the journal style.

We hope that our explanations are clear. Otherwise, don’t hesitate to contact us for further explanation. The first author joins a holiday from 20-27 March.

In anticipation of your reaction,
with kind regards,

Else M. Zantinge
Peter F.M. Verhaak
Dinny H. de Bakker
Klaas van der Meer
Jozien M. Bensing

NIVEL, Netherlands Institute for Health Services Research
Department of General Practice, University of Groningen